Author's response to reviews

Title: Epidemiological investigations of human rabies in China

Authors:

Miao Song (lpssongm@126.com)
Qing Tang (qtang04@sina.com)
Ding-Ming Wang (wangdingm123@sina.com)
Zhao-Jun Mo (mozhi@126.com)
Shou-Heng Guo (gshengq2490@163.com)
Hao Li (haoli11@hotmail.com)
Xiao-Yan Tao (txy212@126.com)
Charles E Rupprecht (cyr5@cdc.gov)
Zi-Jian Feng (fengzj64@sina.com)
Guo-Dong Liang (gdliang@hotmail.com)

Version: 9 Date: 10 December 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

Thanks for your email. In the new version, only the statistical symbols were corrected.

Best regards

Yours sincerely

Miao Song, M.D.